Author's response to reviews

Title: Weight change and incident metabolic syndrome in Iranian men and women; a 3 year follow-up study

Authors:

Azadeh Zabetian (zabetian@endocrine.ac.ir)
Farzad Hadaegh (fzhadaegh@endocrine.ac.ir)
Parvin Sarbakhsh (p.sarbakhsh@gmail.com)
Fereidoun Azizi (azizi@endocrine.ac.ir)

Version: 6 Date: 9 May 2009

Author's response to reviews: see over
Dear Dr. Mark Todd

Assistant Editor of BMC-series journals

Thank you for considering our manuscript: "Weight change and incident metabolic syndrome in Iranian men and women; a 3 year follow-up study", Ms. Ref. NO. 1014196674217928, for publication in the “BMC Public Health” journal; it has been formatted based on the journal comments, on the e-mail dated 06-05-2009, and the new version of the manuscript has been resubmitted to the submission system.

We hope that this format changed manuscript meets the journal’s approval and look forward to hearing from you.

Sincerely,

Farzad Hadaegh
Associate professor of Internal Medicine and Endocrinology
Prevention of Metabolic Disorders Research Center,
Research Institute for Endocrine Sciences,
Shahid Beheshti University (M.C.)
P.O. Box: 19395-4763, Tehran, Islamic Republic of Iran
Phone: +98 21 22432503
Fax: +98 21 22402463
Email: fzhadaegh@endocrine.ac.ir